Author’s response to reviews

**Title:** Prevalence and key radiographic spinal malalignment parameters that influence the risk for gastroesophageal reflux disease in patients treated surgically for adult spinal deformity

**Authors:**

Tetsuro Ohba (tooba@yamanashi.ac.jp)

Shigeto Ebata (sebata@yamanashi.ac.jp)

Kensuke Koyama (kkoyama@yamanashi.ac.jp)

Hirotaka Haro (haro@yamanashi.ac.jp)

**Version:** 3  **Date:** 24 Dec 2017

Author’s response to reviews:

Thank you very much for the excellent reviews. We have addressed each comment, criticism, and suggestion individually below. We believe that the changes suggested by the reviewers have improved this manuscript. We hereby resubmit this document with point-by-point responses to all of the reviewers’ comments along with the updated version of our manuscript, which has been revised in accordance with the comments from the editor.

Editor Comments:

Dear authors

1. Please have the text edited by a professional language editing service or a native English speaking colleague. There are many issues with grammar, wording, spelling, and/or punctuation that need to be addressed.

   → With respect to this suggestion, manuscript has been looked over by professional language editing service and certification form is presented with revised manuscript.
2. Please include the email addresses for all authors on the title page. The corresponding author should still be indicated.

→ We have provided the email addresses for all authors on the title page.

3. Please, amend your Data availability statement. The Availability of data and materials section refers to the raw data used in your study. Data availability statements can take one of the following forms (or a combination of more than one if required for multiple datasets):

→ Now, we have amended our data availability statement according to editor comment.